\documentclass[11pt, oneside]{article}   	% use "amsart" instead of "article" for AMSLaTeX format
\usepackage{geometry}                		% See geometry.pdf to learn the layout options. There are lots.
\geometry{letterpaper}                   		% ... or a4paper or a5paper or ... 
%\geometry{landscape}                		% Activate for for rotated page geometry
%\usepackage[parfill]{parskip}    		% Activate to begin paragraphs with an empty line rather than an indent
\usepackage{graphicx}				% Use pdf, png, jpg, or eps§ with pdflatex; use eps in DVI mode
								% TeX will automatically convert eps --> pdf in pdflatex		
\usepackage{amssymb}

\title{New Assignment 8: Asymptotes}
\author{}
%\date{}							% Activate to display a given date or no date

\begin{document}
\maketitle
%\section{}
%\subsection{}
\noindent{\bf 1)}  Describe the horizontal asymptotes of the following  rational functions; be sure to show complete work.
(a) $f(x)=\frac{3x-1}{2-5x} $ (b) $h(x)=\frac{x+7}{x^2-6x+8}$ (c) $c(x)=\frac{7x^2-3x+2x^3+6}{4x-x^2-2-5x^3}.$

\medskip

\noindent{\bf 2)} Find all vertical asymptotes, horizontal asymptotes, $x$-intercepts and $y$-intercepts for 
a) $f(x)=\frac{3x-1}{2-5x}$ b) $g(x)=\frac{2x}{x+7}$  (c) $h(x)=\frac{x+7}{x^2-6x+8}.$


\end{document}  